Author's response to reviews

Title: The new Lyon brace: ARTbrace. New concepts based on scoliosis detorsion. Prospective study of the first 225 early radiological results of in-brace correction

Authors:

Jean Claude JC de Mauroy (demauroy@aol.com)
Cyril C Lecante (cyril.lecante@lecante.com)
Frédéric F Barral (frederic.barral@lecante.com)
Sophie S Pourret (sophie.pourret@lecante.com)

Version: 4 Date: 7 October 2014

Author's response to reviews:

Added in conclusion
"This first paper is an introduction with very short results and does not prejudge the final outcome."